Author’s response to reviews

Title: The effects of ibuprofen on the physiology and survival on the rabbit endotoxic shock

Authors:

Ilhami Celik (ilhamicelik@hotmail.com)
Ayhan Akbulut (a.akbulut@yahoo.com)
Sirri S Kilic (sskilic@hotmail.com)
Ali Rahman (arahman@hotmail.com)
Pervin Vural (pervinvural@dostmail.com)
Mukaddes Canbaz (mcanbaz@hotmail.com)
Suleyman Felek (feleks@hotmail.com)

Version: 2 Date: 17 May 2002

Thank you for early replying my e-mail. The previously published part of this study was added at references list and it was mentioned at the part of the end of the discussion.